PEER REVIEW HISTORY

BMJ Open publishes all reviews undertaken for accepted manuscripts. Reviewers are asked to complete a checklist review form (http://bmjopen.bmj.com/site/about/resources/checklist.pdf) and are provided with free text boxes to elaborate on their assessment. These free text comments are reproduced below.

ARTICLE DETAILS

TITLE (PROVISIONAL)	Mechanisms of Chiropractic Spinal Manipulative Therapy for
	Patients with Chronic Primary Low Back Pain: Protocol for a
	Mechanistic Randomised Placebo-Controlled Trial
AUTHORS	Gevers-Montoro, Carlos; Ortega-De Mues, Arantxa; Piché, Mathieu

VERSION 1 – REVIEW

REVIEWER		Glissmann Nim, Casper
		University Hospital of Southern Denmark
REVIEW RETURNED		03-Aug-2022
GENERAL COMMENTS	The Me enj pro Ho cor The this pro onl spe	eviewed the paper" Efficacy of Chiropractic Spinal Manipulative erapy in Patients with Chronic Primary Low Back Pain: Protocol for a schanistic Randomised Placebo-Controlled Trial" with great interest. I oyed the read, and it is such an exciting design that potentially can ovide some very needed answers. Also, the trial is registered correctly. wever, I also have some considerations. Please notice that I have mmented on the manuscript text first and then the abstract at the end. ere are many comments, but I believe that with some adjustments, a could be a very well-read paper. Also, I find it essential for all ofessions involved in SMT. Limiting it to chiropractic is redundant and y reduces the interest from other groups. There are some more ecific comments below. Again, thank you for conducting such eresting research. I am very much looking forward to the results.
	Inti	oduction:
su the pa ref cri wa 16 wit sta se A o pa stii jus	sur the pai refe crit wa 163 wit	m having difficulty following the statement on p. 4, I. 15-20. I am not re how subgrouping patients according to pain mechanisms is" One of better studied classification systems," considering that nociplastic n was adopted only a couple of years ago, with most of the erences 5-10 being before this time. Also, recently some have icized how clinically relevant it is, considering that we have no robust y of reliable assessing it (See more at PAIN: August 2022 - Volume 3 - Issue 8 - p e963 DOI: 10.1097/j.pain.000000000002662). I agree h the authors that this is very interesting and a solid point to set the ge, but it does require less certainty, in my opinion. The following ction is more appropriate and could stand by itself.
	puick notion about temporal summation, while it is of interest in the per, examined temporal summation using cuff algometry (a relevant nulus for MSK patients) on chronic LBP patients and found nothing, t a consideration whenever your result is available tps://chiromt.biomedcentral.com/articles/10.1186/s12998-021-00367-	

Also, the same authors failed to find an association between CSI and QST results, and this is an uncommon finding (https://pubmed.ncbi.nlm.nih.gov/34668367/)
These are just considerations. There is definitely a link between altered pain sensitivity, inflammation, and psychology, but at this point, we must consider it speculative. It will be interesting to see your results also in a setting where manual therapy is applied to patients.
The aim is somewhat confusing, and I think it is due to the title. It states" efficacy," and then there is nothing about whether SMT is better than placebo for clinical efficacy, and this is obviously because you are looking at it mechanistically. I would change the title to" The mechanism of Spinal Manipulative Therapy in Patients with Chronic Primary Low Back Pain: Protocol for a Mechanistic Randomised Placebo-Controlled Trial." In this sense, I would also omit the" Chiropractor" part. There is nothing special about chiropractors performing SMT compared to other manual-therapy professions, and it would also allow your results to be more welcoming in other professional circuits. Then I would change the aim to three specific questions using a list format. Objective 1: find a CS phenotype within persistent LBP Objective 2: Which variables are associated with clinical response to SMT compared to a placebo
I can suggest reading this paper on association vs. prediction studies https://bmcmedresmethodol.biomedcentral.com/articles/10.1186/s12874-020-01050-7. Especially Figure 1.
Method:
P6., L37, I would recommend writing" the protocol is reported according to SPIRIT."
I have no comments on the selection criteria or randomization as the study is already enrolling participants (according to the registration). The information is well presented.
Please add what side the patient will be lying on.
Please also state whether the clinician performing the SMT is part of the research team.
Usually, in RCTs, there is one primary outcome and one end-point. This sounds like pain at week 4 or perhaps 12? but please clarify. I would not call disability a primary outcome, as your sample size is based on pain. Instead, it is also a secondary outcome, just a clinical one, that should belong under" clinical examination variables" as part of the five topics.
CLBP trajectory (duration and frequency), how is this measured?
P. 10 at the top. Please elaborate on how clinicians will determine" whether the CLBP is proportionate or disproportionate to the degree or nature of the injury or pathology."
Please see my comment above about the use of CSI, especially using the cut-point of 40 in CLBP.
Generally, it is described that measurements are collected at baseline

and follow-up. Please clarify if this includes all follow-up time points, 4, 8, and 12 weeks.
P.12, I. 31: Perhaps I am misunderstanding but does reference 81 not suggest using the mean of 3 trials?
P. 12, L. 42, what is verified by palpation?
I am not familiar with the process of collecting TNF- α , so I cannot provide any revisions on this aspect.
Going through the method, I was a bit disappointed that temporal summation is not measured, considering its lengthy introduction. I would now suggest removing it altogether in the introduction to make it more relevant to the method actually used.
I enjoyed the section on adverse events; thank you for being so detailed about it.
Procedure:
Will participants in the placebo arm be offered" real" SMT after 12 weeks or after 4?
Also, would you mind providing some more information about MCC? It is comparable to a primary care setting or is a more typical chiropractic school clinic. In addition, the clinicians often referred to throughout the manuscript are these students or chiropractors, and how experienced are they? This is especially important as they are tasked heavily with some very detailed inclusion criteria and having to score patients for a predictor variable.
Could you please clarify your sample size calculation 1: You state five independent variables, but I can read multiple throughout the method section (The PPTs alone are 4, pain site, distant to pain site, leg, and hand). Do you apply some principal component analysis to cut it down to five variables, or do you select the most relevant variables from the five" topics"? If so, this ought to be stated here. Please clarify. Also, have you considered what to do about dropout or loss to follow-up?
For the RCT calculation, how come you did not consider the results from the Rubinstein review on chronic LBP? Here they state a change at one month as -7.55 (-19.86 to 4.76) on a $0 - 100$ scale. Indicating no statistically significant difference. Perhaps your sham provides less mechanical input.
ANOVA should be defined in the text before first use
I would recommend highlighting the five key predictors referred to in the sample size calculation earlier to avoid confusion.
For the statistical analysis section, I would ensure that the proposed objectives (1 to 3) are reported in the same order to avoid confusion. Perhaps even with subheadings. In the objectives, nowhere does it say anything about the clinical efficacy of SMT compared to placebo, and that is not really what is interesting about this study either, but the mechanistic approach. Yet in the statistical analysis, I can primarily read about how the two groups are being compared.
I enjoyed the discussion and would recommend that the method reflects

more along this path rather than SMT vs. placebo. The major strength in this trial is the placebo group, not for clinical outcomes but the mechanistic ones.
P. 19, I 59:" and of the clinician delivering care to the patient's progress." It could be misleading. Please clarify
P. 19, I 60:" This will substantially reduce potential biases." Either it will substantially reduce biases, or it may reduce potential bias. Unfortunately, the clinician is still aware of the treatment, a bias that is impossible to overcome. Also, there is only one clinician, making the generalizability limited. This ought to be mentioned in the limitations.
p. 20: I am unsure how one would understand the following:" Although SMT has been found to be as effective as other frequently used and recommended interventions for CLBP, it fails to outperform a placebo under highly controlled circumstances." Especially when you expect a significant enough effect, you only require 68 participants. I would recommend omitting this aspect and adding the other limitations mentioned.
In figure 2, I would rename the different parts as A, B, C, etc., to avoid confusion having a figure 1-5 in figure 2.
Abstract:
I would reflect on the objectives of the proposed introduction above. Making them more transparent and corresponding to those in the manuscript. If you have too many words, it would be easy to cut down on the initial part of the introduction.
Again in the method. I would state the design and the procedures and then describe the statistics, in short, reflecting the objectives.
In the" strength and limitation" under the keywords
I would remove the ones about bias and add something about the placebo group and the reference values from the healthy participants. These two aspects are new and by far the most interesting.

REVIEWER	Ailliet, Luc VU University Medical Centre Amsterdam
REVIEW RETURNED	29-Aug-2022

GENERAL COMMENTS	The paper states inclusion of patients for this study was to start in November 2021. Is there / are there any specific reasons for this
	significant delay?

REVIEWER	Bond, Bryan. University of Saint Mary, Physical Therapy
REVIEW RETURNED	04-Sep-2022

GENERAL COMMENTS	Thank-you the opportunity to review the manuscript titled, "Efficacy
	of Chiropractic Spinal Manipulative Therapy in Patients with Chronic
	Primary Low Back Pain: Protocol for a Mechanistic Randomised
	Placebo-Controlled Trial". I appreciate the novel scientific premise
	associated with this manuscript. Overall, I appreciate the authors'
	well-written manuscript and sound scientific design. However,
	please consider the following comments:

1. I agree with the authors' statement that the primary limitation of the design lies with the sham or placebo manipulation. I also recognize the challenge of the application of an appropriate sham manipulation. Although the authors cite a valid placebo (Chaibi et al., 2015), it appears that the clinician will apply an external load/force to the participants as or stated in the manuscript "an unintended force is applied bilaterally to the gluteal region". It seems plausible that the "unintended force" applied relatively remote (i.e., gluteal region) to the most painful vertebral segment, may still elicit a neurophysiological response that influences nociception. For example, patients with elbow pain demonstrate PPT changes after cervical manipulation (Fernandez-Carnero, J., et al. (2008). "Immediate hypoalgesic and motor effects after a single cervical spine manipulation in subjects with lateral epicondylalgia." J Manipulative Physiol Ther 31(9): 675-681.), and patients with TMD pain demonstrate PPT changes after cervical mobilization (La Touche, R., et al. (2009). "The effects of manual therapy and exercise directed at the cervical spine on pain and pressure pain sensitivity in patients with myofascial temporomandibular disorders." J Oral Rehabil 36(9): 644-652.). Thus, the sham manipulation may have a "real" effect on the outcome measures. Please address this limitation within the manuscript. Also, it may be helpful for the readers to visualize the "real SMT" and "placebo SMT" (i.e., a figure or appendix/supplement with a video demonstration). Thanks. 2. From the manuscript, "Real SMT will be performed with the patient positioned in the lateral decubitus position, and applying a high-speed, low-amplitude force on each side of the manipulated segment, with the aim of generating at least one joint cavitation (perceptible sound)." and "In case of not perceiving a cavitation or satisfactory joint movement, the SMT will be repeated once at the corresponding side." For clarification, each participant will receive a minimum of 2 thrusts (i.e., one left and one right), and possibly a maximum of 4 thrusts in the absence of cavitation? What if the manipulative attempts do not produce resultant cavitation? Please address these comments within the manuscript. Thanks. 3. What if some of the "healthy volunteers" develop pain during the 4-weeks from baseline to follow-up sessions? How will authors track these changes? How will authors account for this potential confounder? Please address these comments within the manuscript. Thanks.

VERSION 1 – AUTHOR RESPONSE

Reviewer: 1

Dr. Casper Glissmann Nim, University Hospital of Southern Denmark Comments to the Author:

I reviewed the paper" Efficacy of Chiropractic Spinal Manipulative Therapy in Patients with Chronic Primary Low Back Pain: Protocol for a Mechanistic Randomised Placebo-Controlled Trial" with great interest. I enjoyed the read, and it is such an exciting design that potentially can provide some very needed answers. Also, the trial is registered correctly. However, I also have some considerations. Please notice that I have commented on the manuscript text first and then the abstract at the end. There are many comments, but I believe that with some adjustments, this could be a very well-read paper. Also, I find it essential for all professions involved in SMT. Limiting it to chiropractic is redundant and only reduces the interest from other groups. There are some more specific comments below. Again, thank you for conducting such interesting research. I am very much looking forward to the results.

Response: We thank the reviewer for very thoughtful comments and a thorough revision. We have responded to every comment and addressed all of them with significant changes to our manuscript. However, we have kept "Chiropractic" in the title, because it was requested by the ethics committee to emphasize what type of clinicians would be delivering the spinal manipulative therapy (SMT) both in the title and the manuscript.

Introduction:

I am having difficulty following the statement on p. 4, I. 15-20. I am not sure how subgrouping patients according to pain mechanisms is" One of the better studied classification systems," considering that nociplastic pain was adopted only a couple of years ago, with most of the references 5-10 being before this time. Also, recently some have criticized how clinically relevant it is, considering that we have no robust way of reliable assessing it (See more at PAIN: August 2022 - Volume 163 - Issue 8 - p e963 DOI: 10.1097/j.pain.0000000002662). I agree with the authors that this is very interesting and a solid point to set the stage, but it does require less certainty, in my opinion. The following section is more appropriate and could stand by itself.

Response: We agree that this is just one of various proposed classification systems. We are also aware of the existing debate regarding the reliability of the diagnostic criteria for nociplastic pain, and how this term has become controversial. For this reason, we have reduced the certainty in our sentence and decided to use the more traditional concept (though also not completely accepted) of central sensitization, and now only mention nociplastic pain when introducing recent reviews specifically reviewing this concept and the link between both nociplastic pain and CS, and its diagnosis are discussed [1-3]. Changes can be read in pages 4-5:

One of the proposed classification systems stratifies patients into specific subgroups according to pain mechanisms (nociceptive, neuropathic or central sensitisation).[4-9] It has been suggested that a large proportion of CLBP patients presents chronic primary pain, which has been linked to altered nociceptive processing.[1 10] Among the phenomena that may underlie this aberrant processing, central sensitization (CS) is likely the predominant mechanism,[1 11] and its involvement in CLBP deserves further research.[12]

Thus, it is plausible that mechanical pain sensitivity may play an important role in defining a CS phenotype in CLBP.[2]

A quick notion about temporal summation, while it is of interest in the paper, examined temporal summation using cuff algometry (a relevant stimulus for MSK patients) on chronic LBP patients and found nothing, just a consideration whenever your result is available (https://chiromt.biomedcentral.com/articles/10.1186/s12998-021-00367-4)

Also, the same authors failed to find an association between CSI and QST results, and this is an uncommon finding (https://pubmed.ncbi.nlm.nih.gov/34668367/)

These are just considerations. There is definitely a link between altered pain sensitivity, inflammation, and psychology, but at this point, we must consider it speculative. It will be interesting to see your results also in a setting where manual therapy is applied to patients.

Response: We are aware of both articles. The findings from Holm et al. are relevant to our protocol. CSI and QST evaluate different constructs and we are examining which of these variables/constructs is more closely associated with the response to SMT. As discussed below, we have limited our emphasis on temporal summation, because we don't measure temporal summation in the protocol. We agree that it should be mentioned for its relevance to CS mechanisms. We revised the introduction as follows:

- Page 4, second paragraph:

It was proposed that the pain relieving effects of SMT partly rely on segmental pain inhibition processes.[13] These processes influence temporal summation of pain,[14 15] primary, and secondary hyperalgesia,[16 17] which may be measured to identify patients with a CS phenotype.

- Page 6, first paragraph

The classification of mechanism-based pain phenotypes is a complex and controversial task,[2 18 19] for which a variety of clinical, inflammatory, psychological, and psychophysical constructs must be considered.[8 20] Although CS may influence changes in pain sensitivity induced by SMT,[21] pain phenotyping has been scarcely applied to manual therapy research.[22]

The aim is somewhat confusing, and I think it is due to the title. It states" efficacy," and then there is nothing about whether SMT is better than placebo for clinical efficacy, and this is obviously because you are looking at it mechanistically. I would change the title to" The mechanism of Spinal Manipulative Therapy in Patients with Chronic Primary Low Back Pain: Protocol for a Mechanistic Randomised Placebo-Controlled Trial." In this sense, I would also omit the" Chiropractor" part. There is nothing special about chiropractors performing SMT compared to other manual-therapy professions, and it would also allow your results to be more welcoming in other professional circuits. Then I would change the aim to three specific questions using a list format.

Objective 1: find a CS phenotype within persistent LBP

Objective 2: Which variables are associated with clinical response to SMT compared to a placebo Objective 3: Prediction of clinical response using the variables

Response: We appreciate the reviewer's insight. We agree with the idea of generalizability but in this case specifically, as mentioned above, we must keep the word "Chiropractic" in the title as SMT was delivered by a chiropractor in a chiropractic setting and the Ethics Review Board had requested that this had to be very clearly stated; for Spanish regulations, SMT is not a regulated intervention. However, we revised the title as follows:

Mechanisms of Chiropractic Spinal Manipulative Therapy for Patients with Chronic Primary Low Back Pain: Protocol for a Mechanistic Randomised Placebo-Controlled Trial

Regarding the objectives, we have only two, with an overarching aim. This was clarified as follows (page 6):

The aim of this clinical trial is to investigate whether variables associated with a CS phenotype may help predict the response to SMT. The specific objectives are: 1) to identify the clinical, psychological, psychophysical and inflammatory variables linked to CS in a cohort of CLBP patients; and 2) to examine which of these variables predict the clinical response to SMT.

I can suggest reading this paper on association vs. prediction studies https://bmcmedresmethodol.biomedcentral.com/articles/10.1186/s12874-020-01050-7. Especially Figure 1.

Response: We understand that prediction requires a prospective experimental design, which is what we are using in this protocol. We also agree that a more robust predictive model needs to be tested by using clinical prediction rules first, and then confirmed in RCT's. We hope that our clinical trial results will be helpful in a design of clinical prediction rules for SMT in the future, including variables linked to pain mechanisms phenotypes.

Method:

P6., L37, I would recommend writing" the protocol is reported according to SPIRIT."

Response: The text was revised as follows on page 6:

This protocol is reported according to the guidelines for clinical trial protocols Standard Protocol Items: Recommendations for Interventional Trials[23] (SPIRIT statement).

I have no comments on the selection criteria or randomization as the study is already enrolling participants (according to the registration). The information is well presented.

Please add what side the patient will be lying on.

Response: Patients will receive two SM, one on each side. Therefore, we have revised the text accordingly to clarify (page 8, last paragraph):

Two real SMT will be performed with the patient positioned in the lateral decubitus position (once on each side), by applying a high-velocity, low-amplitude force on the manipulated segment, with the aim of generating at least one joint cavitation (associated with an audible sound).

Please also state whether the clinician performing the SMT is part of the research team.

Response: The information was added on page 8:

Both real and placebo SMT will be delivered by a chiropractor with twenty years of experience that is part of the research team (CG-M).

Usually, in RCTs, there is one primary outcome and one end-point. This sounds like pain at week 4 or perhaps 12? but please clarify. I would not call disability a primary outcome, as your sample size is based on pain. Instead, it is also a secondary outcome, just a clinical one, that should belong under" clinical examination variables" as part of the five topics.

Response: There are RCT's with multiple coprimary outcomes. One good example is a recent RCT by Goertz and colleagues, precisely on the use of chiropractic care for LBP [24]. Nonetheless, we also think that the reviewer makes an excellent point regarding disability, also because our RCT is mostly focused on pain-related mechanisms. For this reason, we have included only pain intensity as a primary outcome, and disability as a secondary outcome measure. In addition, we have clarified that our primary endpoint is at the end of the 12 sessions of SMT, and that there are two additional follow-ups 4 and 12 weeks after the end of treatment. Please refer to page 9, second paragraph for changes:

Average pain intensity will be used as the primary outcome for all statistical analyses. The primary endpoint will be the change from baseline at the completion of the 12 sessions of SMT. For the follow-up, average pain intensity will be assessed 4 and 12 weeks after the completion of the trial.

Also, please not that throughout the text, the terminology has been standardized and adapted to these changes. We now use 'primary outcome' or 'primary variable' to refer exclusively to pain intensity, and 'primary endpoint' to refer to the change in baseline pain intensity at the completion of the 12 treatment sessions.

CLBP trajectory (duration and frequency), how is this measured?

Response: We thank the reviewer for pointing this out. Pain duration will be measured in months since the onset of the first LBP episode, while frequency of pain will be determined by using previously defined pain trajectories. Kongsted and colleagues found clear support to distinguish between two types of pain trajectories (episodic and fluctuating) that may be useful in the detection of CLBP subgroups.[25] This has been specified in the manuscript (page 10, second paragraph):

The duration of CLBP will be calculated as the number of months since the onset of the first episode of LBP. As for pain frequency, participants' CLBP trajectory will be classified as either fluctuating or episodic, depending on whether they recall asymptomatic periods of at least 4 weeks (episodic) or not (fluctuating).[25]

P. 10 at the top. Please elaborate on how clinicians will determine" whether the CLBP is proportionate or disproportionate to the degree or nature of the injury or pathology."

Response: We will be following the criteria originally proposed by Smart et al., where "disproportionate, non-mechanical unpredictable pattern of pain provocation in response to multiple/non-specific aggravating/easing factors" were found to be the best predictors of CS pain.[4] These criteria were later reviewed and adopted by the guidelines published by Nijs and colleagues,[5] who recently reviewed and compared them to IASP criteria to diagnose nociplastic pain.[2] In line with Shraim and colleagues' recent classification system, these criteria will be assessed by subjective pain ratings, and, when available, complemented with information from diagnostic imaging.[8] See the bottom of page 10 for changes:

A diffuse rather than a discrete pain distribution was identified as a key criterion of a CS phenotype.[1 4] Also, classifying symptoms as proportionate (or not) was proposed to differentiate nociceptive pain from CS mechanisms.[2] The pattern of pain distribution and the provocation and response to aggravating and palliative factors will be assessed during case history and physical examination. This will be complemented with information provided by diagnostic imaging when available.[8]

Please see my comment above about the use of CSI, especially using the cut-point of 40 in CLBP.

Response: We understand the reviewer's concern regarding the cutoff score of 40 for the CSI. The CSI was not designed to be used single-handedly, but rather in combination with other diagnostic tools, including for CLBP [5 7]. We understand that the cutoff value may not parallel changes in pain sensitivity (or other constructs linked to CS). Instead, we consider the CSI to be one construct among others that may determine a (multifactorial) CS-phenotype. We revised the manuscript section describing this outcome measure (page 11, third paragraph):

When combined with a clinical presentation suggestive of CS,[2] the CSI is an useful tool to identify patients compatible with certain CS mechanisms, particularly when using the cut-off value of 40 points.[26]

Generally, it is described that measurements are collected at baseline and follow-up. Please clarify if this includes all follow-up time points, 4, 8, and 12 weeks.

Response: As mentioned above, this was clarified both in the manuscript and in Figure 2.

- Page 9:

Average pain intensity will be used as the primary outcome for all statistical analyses. The primary endpoint will be the change from baseline at the completion of the 12 sessions of SMT. For the follow-up, average pain intensity will be assessed 4 and 12 weeks after the completion of the trial.

- Page 11:

The main secondary outcome will be the disability caused by CLBP. After completing the case history, patients will fill out the Oswestry low back disability index questionnaire.[27] The questionnaire will also be completed after the 12th treatment session with the primary endpoint, and at subsequents 4-and 12-week follow-ups.

The Pain Catastrophizing Scale (PCS) and CSI[28 29] will be completed before the beginning of the treatment (baseline) and at a single follow-up after the 12th treatment session.

The scores in these questionnaires will be measured both at baseline and after the 12th treatment session for exploratory purposes. We will examine whether these variables are associated with the primary outcome.

- Pages 12

PPTs will be assessed during the initial examination for baseline, and after the final SMT session (see Figure 2E).

- Page 13:

Before initiating the first treatment session and on the day of the last treatment session, urine samples will be collected (first morning micturition) and stored at -20° C (see Figure 2B and 2F).

- Page 15:

All outcome measures will be re-assessed at the 12th and last treatment session (i.e., the primary endpoint). After completing data collection at the primary endpoint, patients allocated to the placebo arm will be offered the possibility of receiving the "real" SMT, free of charge, at the MCC. In addition, all patients will be contacted for the follow-up of CLBP intensity and disability, 4 and 12 weeks after

the primary endpoint (Figure 2G). Meanwhile, healthy volunteers will participate in two visits (baseline and follow-up after 4 weeks) when all relevant outcomes will be assessed (Figure 3). The study will have a total estimated duration of one year.

P.12, I. 31: Perhaps I am misunderstanding but does reference 81 not suggest using the mean of 3 trials?

Response: In another study, the mean of two or three trials was found to be equally reliable in asymptomatic participants. The same authors reported that one trial may be as reliable as two or three measurements in both asymptomatic and CLBP patients. Considering that we will test both population types, we decided to perform two trials. This is clarified in the revised manuscript (page 12):

Two consecutive measurements provide excellent reliability when assessing both populations with and without LBP,[30 31] while performing two repetitions per side of the lower back was proposed to optimize inter-session reliability.[32]

P. 12, L. 42, what is verified by palpation?

Response: Palpation will be used to verify that the segment indicated by the patient either reproduces clinical pain or is located as close as possible to the area of CLBP symptoms. We intend to select the segment most closely related to clinical pain rather than the segment where pain sensitivity is greater[17]. See page 12 for revisions.

Manual palpation will be performed to confirm that the selected segment either reproduces clinical pain or is the closest to the area (or to the centre) of CLBP symptoms. This will allow to assess the area of primary pain or hyperalgesia (segmental sensitivity).

I am not familiar with the process of collecting TNF- α , so I cannot provide any revisions on this aspect.

Going through the method, I was a bit disappointed that temporal summation is not measured, considering its lengthy introduction. I would now suggest removing it altogether in the introduction to make it more relevant to the method actually used.

Response: It is essential to understand the potential implication of central sensitization mechanisms in studies that examine the mechanisms of SMT. Therefore, we reduced the emphasis on temporal summation, but we did not remove it completely. However, we agree to focus on hyperalgesia. The introduction now reads as follows (page 4, second paragraph):

It was proposed that the pain relieving effects of SMT partly rely on segmental pain inhibition processes[13]. These processes influence temporal summation of pain[14 15], primary, and secondary hyperalgesia,[16 17] which may be measured to identify patients with a CS phenotype.

I enjoyed the section on adverse events; thank you for being so detailed about it.

Procedure:

Will participants in the placebo arm be offered" real" SMT after 12 weeks or after 4?

Response: Yes, as mentioned above. This was clarified on page 15:

After completing data collection at the primary endpoint, patients allocated to the placebo arm will be offered the possibility of receiving the "real" SMT, free of charge, at the MCC. In addition, all patients will be contacted for the follow-up of CLBP intensity and disability, 4 and 12 weeks after the primary endpoint (Figure 2G).

Also, would you mind providing some more information about MCC? It is comparable to a primary care setting or is a more typical chiropractic school clinic.

Response: This is an important detail. We added the following to page 6:

The MCC clinic is a primary care setting specialized in spine care, including chiropractic and physical therapy services.

In addition, the clinicians often referred to throughout the manuscript are these students or chiropractors, and how experienced are they? This is especially important as they are tasked heavily with some very detailed inclusion criteria and having to score patients for a predictor variable.

Response: In order to avoid confusion, we avoided the use of the word clinician in the methods section. Instead, we used the term "investigator" and specified their task (i.e., delivering care), while the investigators assessing the patient are designated as "outcome assessors". The outcome assessors are graduate students with training and expertise in the measurement of PPTs, as described on page 12:

PPTs will be assessed by two interns completing their Master's in Chiropractic degree, after three months of training and pilot data collection. One of the two outcome assessors will be randomly assigned to each patient to perform both baseline and follow-up measurements.

The inclusion of participants is the responsibility of the investigator delivering care, as specified on page 7:

An investigator with over twenty years of clinical experience will be responsible for the selection of participants.

Could you please clarify your sample size calculation 1: You state five independent variables, but I can read multiple throughout the method section (The PPTs alone are 4, pain site, distant to pain site, leg, and hand). Do you apply some principal component analysis to cut it down to five variables, or do you select the most relevant variables from the five" topics"? If so, this ought to be stated here. Please clarify. Also, have you considered what to do about dropout or loss to follow-up?

Response: We revised the statistical analysis section to clarify this issue and introduce the predictor variables early in the section on page 15. We also revised other manuscript sections, as discussed in one of the reviewer's comments below.

A multiple regression analysis will be performed using five independent variables described in the statistical analysis section as predictors. These variables include baseline values of local PPTs, urinary concentrations of TNF, scores in PCS and CSI questionnaires and a priori expectations of pain relief.

Response: Regarding dropout and loss to follow-up, we have provided this information on page 16:

A total of 110 patients will be recruited, accounting for an estimated dropout rate of 5-10%.

For the RCT calculation, how come you did not consider the results from the Rubinstein review on chronic LBP? Here they state a change at one month as -7.55 (-19.86 to 4.76) on a 0 - 100 scale. Indicating no statistically significant difference. Perhaps your sham provides less mechanical input.

Response: It is more conservative and careful to base our power calculation on the detection of a theoretical small effect in order to have sufficient statistical power to reach a conclusion with data from in this clinical trial. We are more confident that we will not miss an effect and report a false negative. The values in one study may be an indicator, but the effects vary from study to study.

ANOVA should be defined in the text before first use

Response: The acronym was defined on page 16, when it is now first introduced:

A series of mixed analyses of variance (ANOVA) will be performed to examine differences in PPTs, urinary TNF- α levels, PCS, CSI, BDI-II and GAD scores at baseline and primary endpoint between the three groups (control, SMT and placebo).

I would recommend highlighting the five key predictors referred to in the sample size calculation earlier to avoid confusion.

Response: We revised the text on page 10 to clarify, and page 15:

For the present study, all categories will be considered except the last one, which will only be used to rule out pain of suspected neuropathic or nociceptive aetiology. Variables belonging to these categories will be assessed for exploratory purposes and five of them will be examined as predictors of the response to SMT (two questionnaires, one quantitative sensory testing variable, one laboratory test variable and the expectations of pain relief).

A multiple regression analysis will be performed using five independent variables described in the statistical analysis section as predictors. These variables include baseline values of local PPTs, urinary concentrations of TNF, scores in PCS and CSI questionnaires and a priori expectations of pain relief.

For the statistical analysis section, I would ensure that the proposed objectives (1 to 3) are reported in the same order to avoid confusion. Perhaps even with subheadings. In the objectives, nowhere does it say anything about the clinical efficacy of SMT compared to placebo, and that is not really what is interesting about this study either, but the mechanistic approach. Yet in the statistical analysis, I can primarily read about how the two groups are being compared.

Response: The text was revised to clarify (see pages 16-18).

I enjoyed the discussion and would recommend that the method reflects more along this path rather than SMT vs. placebo. The major strength in this trial is the placebo group, not for clinical outcomes but the mechanistic ones.

Response: The revisions made according to comments above strengthened the emphasis on the predictor variables, which is in line with this comment.

P. 19, I 59:" and of the clinician delivering care to the patient's progress." It could be misleading. Please clarify

Response: The text was revised to clarify on page 20, second paragraph:

Moreover, the investigator delivering care will be blinded to the patients' progress.

P. 19, I 60:" This will substantially reduce potential biases." Either it will substantially reduce biases, or it may reduce potential bias. Unfortunately, the clinician is still aware of the treatment, a bias that is impossible to overcome. Also, there is only one clinician, making the generalizability limited. This ought to be mentioned in the limitations.

Response: This was clarified on pages 20-21 as follows:

This will reduce biases that are typically introduced in manual therapy trials.

(...)

Regarding potential limitations, having only one clinician may limit the generalizability of the SMT effects. However, it also has the advantage of standardizing the interventions and reducing variability in the procedures. It should also be noted that although blinding the investigator providing care is desirable, it is impossible in manual therapy trials[33], including the present study.

p. 20: I am unsure how one would understand the following:" Although SMT has been found to be as effective as other frequently used and recommended interventions for CLBP, it fails to outperform a

placebo under highly controlled circumstances." Especially when you expect a significant enough effect, you only require 68 participants. I would recommend omitting this aspect and adding the other limitations mentioned.

Response: The manuscript was revised including the limitations suggested by the reviewer, as detailed above. Also, we included all limitations in one paragraph (page 21):

It should also be noted that although blinding the investigator providing care is desirable, it is impossible in manual therapy trials[33], including the present study. As the sham and real SMT have a high degree of similarity, effective blinding of participants is feasible.[34] The inability to distinguish the placebo from the real treatment is desirable to limit interpretation bias, particularly in a mechanistic trial as in the present study.[35] However, the sham SMT may rely on specific mechanisms that overlap with those of real SMT, leading to treatment effects.[35 36] Accordingly, the sham SMT should not be considered as an inert placebo and the lack of between-group differences should be interpreted with caution, with a potential risk for type II errors.

In figure 2, I would rename the different parts as A, B, C, etc., to avoid confusion having a figure 1-5 in figure 2.

Response: During the revision process, Figure 2 was revised and we included a new Figure 3

See revised version of Figure 2, and new Figure 3

Abstract:

I would reflect on the objectives of the proposed introduction above. Making them more transparent and corresponding to those in the manuscript. If you have too many words, it would be easy to cut down on the initial part of the introduction.

Response: We agree with the reviewers that the objectives need to be presented according to the changes made in the manuscript. The abstract was revised as follows:

This protocol describes a randomised placebo-controlled trial aiming to examine which variables linked to central sensitisation may help predict the clinical response to SMT in a cohort of CLBP patients.

Again in the method. I would state the design and the procedures and then describe the statistics, in short, reflecting the objectives.

Response: The abstract was revised as follows:

Pain intensity will be assessed as the primary outcome after completing the 4-week treatment (primary endpoint), and at 4- and 12-week follow-ups. Baseline values of two pain questionnaires, lumbar pressure pain thresholds, concentrations of an inflammatory cytokine and expectations of pain relief will be entered as predictors of the response to SMT in a multiple regression model. Changes in these variables after treatment will also be used in a second multiple regression model. The reference values of these predictors will be measured from 50 age and sex-matched healthy controls to allow interpretation of values in patients. Mixed analyses of variance will also be conducted to compare the primary and secondary outcome measures between groups (SMT vs. placebo) over time (baseline vs. post-treatment).

In the" strength and limitation" under the keywords

I would remove the ones about bias and add something about the placebo group and the reference values from the healthy participants. These two aspects are new and by far the most interesting.

Response: we have also updated the strengths and limitations to reflect the suggestions made by the reviewer and the revised manuscript:

• The design including a control group with healthy participants will allow confirming the usefulness of a classification system for patients with chronic primary low back pain according to the underlying

pain mechanisms.

• The blinding of outcome assessors, statistician, laboratory technician, and of the investigator providing care to the patients' progress will contribute to reduce bias.

A high degree of similarity between the sham and real manipulations increases the odds of successfully blinding participants. However, the sham intervention may produce clinical effects.
Clinical trials on manual therapy, including the present study, are limited by the impossibility of blinding the investigator providing care to the intervention.
Reviewer: 2

Dr. Luc Ailliet, VU University Medical Centre Amsterdam

Comments to the Author:

The paper states inclusion of patients for this study was to start in November 2021. Is there / are there any specific reasons for this significant delay?

Response: The delays at the beginning of the study are related to restrictions and other measures to mitigate the impact of the pandemic. Data collection was sometimes completely halted for over a month during because of surges in COVID-19 cases and hospitalizations. Data collection is currently ongoing and is planned to be completed in 2023.

Reviewer: 3 Dr. Bryan. Bond, University of Saint Mary, Allen College

Comments to the Author:

Thank-you the opportunity to review the manuscript titled, "Efficacy of Chiropractic Spinal Manipulative Therapy in Patients with Chronic Primary Low Back Pain: Protocol for a Mechanistic Randomised Placebo-Controlled Trial". I appreciate the novel scientific premise associated with this manuscript. Overall, I appreciate the authors' well-written manuscript and sound scientific design. However, please consider the following comments:

1. I agree with the authors' statement that the primary limitation of the design lies with the sham or placebo manipulation. I also recognize the challenge of the application of an appropriate sham manipulation. Although the authors cite a valid placebo (Chaibi et al., 2015), it appears that the clinician will apply an external load/force to the participants as or stated in the manuscript "an unintended force is applied bilaterally to the gluteal region". It seems plausible that the "unintended force" applied relatively remote (i.e., gluteal region) to the most painful vertebral segment, may still elicit a neurophysiological response that influences nociception. For example, patients with elbow pain demonstrate PPT changes after cervical manipulation (Fernandez-Carnero, J., et al. (2008). "Immediate hypoalgesic and motor effects after a single cervical spine manipulation in subjects with lateral epicondylalgia." J Manipulative Physiol Ther 31(9): 675-681.), and patients with TMD pain demonstrate PPT changes after cervical mobilization (La Touche, R., et al. (2009). "The effects of manual therapy and exercise directed at the cervical spine on pain and pressure pain sensitivity in patients with myofascial temporomandibular disorders." J Oral Rehabil 36(9): 644-652.). Thus, the sham manipulation may have a "real" effect on the outcome measures. Please address this limitation within the manuscript. Also, it may be helpful for the readers to visualize the "real SMT" and "placebo SMT" (i.e., a figure or appendix/supplement with a video demonstration). Thanks.

Response: We share the reviewer's interest in the placebo mechanisms of SMT (see section 2.4 in Gevers-Montoro et al, Eur J Pain, 2021[36]). The placebo SMT chosen for this study was reported to blind participants adequately[36]. However, as the reviewer suggests, this procedure may produce clinical effects. This is addressed in the limitation section of the revised manuscript (see the new paragraph in the limitations section, page 21). We have included a supplementary figure S1 were both the placebo and real SMT procedures are illustrated as per the reviewer's suggestion.

It should also be noted that although blinding the investigator providing care is desirable, it is impossible in manual therapy trials[33], including the present study. As the sham and real SMT have a high degree of similarity, effective blinding of participants is feasible.[34] The inability to distinguish the placebo from the real treatment is desirable to limit interpretation bias, particularly in a

mechanistic trial as in the present study.[35] However, the sham SMT may rely on specific mechanisms that overlap with those of real SMT, leading to treatment effects.[35 36] Accordingly, the sham SMT should not be considered as an inert placebo and the lack of between-group differences should be interpreted with caution, with a potential risk for type II errors.

2. From the manuscript, "Real SMT will be performed with the patient positioned in the lateral decubitus position, and applying a high-speed, low-amplitude force on each side of the manipulated segment, with the aim of generating at least one joint cavitation (perceptible sound)." and "In case of not perceiving a cavitation or satisfactory joint movement, the SMT will be repeated once at the corresponding side." For clarification, each participant will receive a minimum of 2 thrusts (i.e., one left and one right), and possibly a maximum of 4 thrusts in the absence of cavitation? What if the manipulative attempts do not produce resultant cavitation? Please address these comments within the manuscript. Thanks.

Response: We thank the reviewer for this important comment. There will be a maximum of 2 attempts to obtain a cavitation on each side (total of 4) and the number of cavitations will be documented (0, 1 or 2) for SMT and placebo sessions. It is unclear whether joint cavitation has any influence in patient outcomes, which could explain clinical equivalence between manipulation and mobilization techniques[37]. However, on exploratory analyses, the number of cavitations will be used as a covariate to examine whether it has any impact on the outcomes. We clarified these issues on pages 8-9:

For this, the chiropractor will use the hypothenar surface or the last phalanx of the 2nd and / or 3rd fingers of the hand to contact the spinous process of the vertebral segment with the most intense clinical pain, as detected in the initial patient examination (see supplemental Figure S1A). In case of not perceiving a cavitation or satisfactory joint movement, SMT may be repeated once on each side. Therefore, all participants will receive a minimum of two and a maximum of four SMT thrusts. Participants in the placebo arm will receive a validated sham SMT that is very similar to SMT.[34] The patient is positioned in the same lateral decubitus position, with the lower leg in extension and the upper leg in flexion, and an unintended force is applied bilaterally to the gluteal region (Figure S1B). The number of real or placebo SMT attempts resulting in joint cavitation will be recorded.

3. What if some of the "healthy volunteers" develop pain during the 4-weeks from baseline to followup sessions? How will authors track these changes? How will authors account for this potential confounder? Please address these comments within the manuscript. Thanks.

Response: This is a very good point. Any healthy participant experiencing acute pain or other symptoms of > 7 days of duration, being injured, initiating any new treatment or receiving a diagnosis compatible with the exclusion criteria for this group will be excluded from the study. If the participant experiences pain or is taking pain medication < 24 hours prior to the follow-up session, this session will be postponed for a maximum of one week. Please see page 14:

Healthy volunteers will be contacted one week prior to the follow-up appointment to rule out any of the following criteria that would exclude them from the follow-up: presence of pain or other symptoms for > 7 days, trauma or injury, initiating a new treatment or receiving a diagnosis compatible with the exclusion criteria. In addition, if the participant reports any pain or taking any pain medication within 24 hours of the follow-up, this session will be postponed for up to one week.

References

1. Kosek E, Clauw D, Nijs J, et al. Chronic nociplastic pain affecting the musculoskeletal system: clinical criteria and grading system. Pain 2021 doi: 10.1097/j.pain.00000000002324 [published Online First: 2021/05/12]

2. Nijs J, Lahousse A, Kapreli E, et al. Nociplastic Pain Criteria or Recognition of Central Sensitization? Pain Phenotyping in the Past, Present and Future. J Clin Med 2021;10(15) doi: 10.3390/jcm10153203 [published Online First: 2021/08/08]

3. Graven-Nielsen T. Mechanisms and manifestations in musculoskeletal pain: from experimental to clinical pain settings. Pain 2022 doi: 10.1097/j.pain.000000000002690 [published Online First: 2022/08/20]

4. Smart KM, Blake C, Staines A, et al. Mechanisms-based classifications of musculoskeletal pain:

part 1 of 3: symptoms and signs of central sensitisation in patients with low back (+/- leg) pain. Manual therapy 2012;17(4):336-44. doi: 10.1016/j.math.2012.03.013 [published Online First: 2012/04/27]

5. Nijs J, Apeldoorn A, Hallegraeff H, et al. Low back pain: guidelines for the clinical classification of predominant neuropathic, nociceptive, or central sensitization pain. Pain Physician 2015;18(3):E333-46. [published Online First: 2015/05/23]

6. O'Sullivan P, Waller R, Wright A, et al. Sensory characteristics of chronic non-specific low back pain: a subgroup investigation. Manual therapy 2014;19(4):311-8. doi: 10.1016/j.math.2014.03.006 [published Online First: 2014/04/16]

7. Nijs J, George S, Clauw D, et al. Central sensitisation in chronic pain conditions: latest discoveries and their potential for precision medicine. The Lancet Rheumatology 2021

8. Shraim MA, Masse-Alarie H, Hodges PW. Methods to discriminate between mechanism-based categories of pain experienced in the musculoskeletal system: a systematic review. Pain

2021;162(4):1007-37. doi: 10.1097/j.pain.00000000002113 [published Online First: 2020/11/03] 9. Vardeh D, Mannion RJ, Woolf CJ. Toward a Mechanism-Based Approach to Pain Diagnosis. J Pain 2016;17(9 Suppl):T50-69. doi: 10.1016/j.jpain.2016.03.001 [published Online First: 2016/09/03] 10. Nicholas M, Vlaeyen JWS, Rief W, et al. The IASP classification of chronic pain for ICD-11: chronic primary pain. Pain 2019;160(1):28-37. doi: 10.1097/j.pain.0000000000001390 [published Online First: 2018/12/27]

 Shraim MA, Masse-Alarie H, Hall LM, et al. Systematic Review and Synthesis of Mechanismbased Classification Systems for Pain Experienced in the Musculoskeletal System. Clin J Pain 2020;36(10):793-812. doi: 10.1097/AJP.000000000000860 [published Online First: 2020/08/28]
 den Bandt HL, Paulis WD, Beckwee D, et al. Pain Mechanisms in Low Back Pain: A Systematic Review With Meta-analysis of Mechanical Quantitative Sensory Testing Outcomes in People With Nonspecific Low Back Pain. The Journal of orthopaedic and sports physical therapy 2019;49(10):698-715. doi: 10.2519/jospt.2019.8876 [published Online First: 2019/08/25]

13. Gevers-Montoro C, Provencher B, Descarreaux M, et al. Neurophysiological mechanisms of chiropractic spinal manipulation for spine pain. Eur J Pain 2021 doi: 10.1002/ejp.1773 [published Online First: 2021/04/01]

14. Randoll C, Gagnon-Normandin V, Tessier J, et al. The mechanism of back pain relief by spinal manipulation relies on decreased temporal summation of pain. Neuroscience 2017;349:220-28. doi: 10.1016/j.neuroscience.2017.03.006 [published Online First: 2017/03/16]

15. Bialosky JE, Bishop MD, Robinson ME, et al. Spinal manipulative therapy has an immediate effect on thermal pain sensitivity in people with low back pain: a randomized controlled trial. Phys Ther 2009;89(12):1292-303. doi: 10.2522/ptj.20090058 [published Online First: 2009/10/03]

16. Gevers-Montoro C, Provencher B, Northon S, et al. Chiropractic Spinal Manipulation Prevents Secondary Hyperalgesia Induced by Topical Capsaicin in Healthy Individuals. Front Pain Res (Lausanne) 2021;2:702429. doi: 10.3389/fpain.2021.702429 [published Online First: 2022/03/18]

17. Nim CG, Kawchuk GN, Schiottz-Christensen B, et al. The effect on clinical outcomes when targeting spinal manipulation at stiffness or pain sensitivity: a randomized trial. Sci Rep

2020;10(1):14615. doi: 10.1038/s41598-020-71557-y [published Online First: 2020/09/05] 18. Hoegh M, Schmid AB, Hansson P, et al. Not being able to measure what is important, does not make things we can measure important. Pain 2022;163(8):e963. doi:

10.1097/j.pain.000000000002662 [published Online First: 2022/07/16]

19. Shraim MA, Sluka KA, Sterling M, et al. Features and methods to discriminate between mechanism-based categories of pain experienced in the musculoskeletal system: a Delphi expert consensus study. Pain 2022;163(9):1812-28. doi: 10.1097/j.pain.00000000002577 [published Online First: 2022/03/24]

20. Holm LA, Nim CG, Lauridsen HH, et al. "Convergent validity of the central sensitization inventory and experimental testing of pain sensitivity". Scand J Pain 2022;22(3):597-613. doi: 10.1515/sjpain-2021-0090 [published Online First: 2021/10/21]

21. Nim CG, Weber KA, Kawchuk GN, et al. Spinal manipulation and modulation of pain sensitivity in persistent low back pain: a secondary cluster analysis of a randomized trial. Chiropr Man Therap 2021;29(1):10. doi: 10.1186/s12998-021-00367-4 [published Online First: 2021/02/26]

22. Damian K, Chad C, Kenneth L, et al. Time to evolve: the applicability of pain phenotyping in manual therapy. The Journal of manual & manipulative therapy 2022;30(2):61-67. doi: 10.1080/10669817.2022.2052560 [published Online First: 2022/03/29]

23. Chan AW, Tetzlaff JM, Gotzsche PC, et al. SPIRIT 2013 explanation and elaboration: guidance for protocols of clinical trials. BMJ 2013;346:e7586. doi: 10.1136/bmj.e7586 [published Online First: 2013/01/11]

24. Goertz CM, Long CR, Vining RD, et al. Effect of Usual Medical Care Plus Chiropractic Care vs Usual Medical Care Alone on Pain and Disability Among US Service Members With Low Back Pain: A Comparative Effectiveness Clinical Trial. JAMA Netw Open 2018;1(1):e180105. doi: 10.1001/jamanetworkopen.2018.0105 [published Online First: 2019/01/16]

25. Kongsted A, Hestbaek L, Kent P. How can latent trajectories of back pain be translated into defined subgroups? BMC musculoskeletal disorders 2017;18(1):285. doi: 10.1186/s12891-017-1644-8 [published Online First: 2017/07/05]

26. Scerbo T, Colasurdo J, Dunn S, et al. Measurement Properties of the Central Sensitization Inventory: A Systematic Review. Pain Pract 2018;18(4):544-54. doi: 10.1111/papr.12636 [published Online First: 2017/08/30]

27. Alcántara-Bumbiedro S, Flórez-García M, Echávarri-Pérez C, et al. Escala de incapacidad por dolor lumbar de Oswestry. Rehabilitación 2006;40(3):150-58.

28. Garcia Campayo J, Rodero B, Alda M, et al. [Validation of the Spanish version of the Pain Catastrophizing Scale in fibromyalgia]. Med Clin (Barc) 2008;131(13):487-92. doi: 10.1157/13127277 [published Online First: 2008/11/15]

29. Cuesta-Vargas AI, Roldan-Jimenez C, Neblett R, et al. Cross-cultural adaptation and validity of the Spanish central sensitization inventory. Springerplus 2016;5(1):1837. doi: 10.1186/s40064-016-3515-4 [published Online First: 2016/11/08]

30. Balaguier R, Madeleine P, Vuillerme N. Intra-session absolute and relative reliability of pressure pain thresholds in the low back region of vine-workers: ffect of the number of trials. BMC musculoskeletal disorders 2016;17(1):350. doi: 10.1186/s12891-016-1212-7 [published Online First: 2016/08/20]

 Balaguier R, Madeleine P, Vuillerme N. Is One Trial Sufficient to Obtain Excellent Pressure Pain Threshold Reliability in the Low Back of Asymptomatic Individuals? A Test-Retest Study. PloS one 2016;11(8):e0160866. doi: 10.1371/journal.pone.0160866 [published Online First: 2016/08/12]
 Liew B, Lee HY, Rugamer D, et al. A novel metric of reliability in pressure pain threshold measurement. Sci Rep 2021;11(1):6944. doi: 10.1038/s41598-021-86344-6 [published Online First: 2021/03/27]

33. Hohenschurz-Schmidt D, Draper-Rodi J, Vase L, et al. Blinding and Sham Control Methods in Trials of Physical, Psychological, and Self-Management Interventions for Pain (Article I): a Systematic Review and Description of Methods. PAIN 2022:10.1097.

34. Chaibi A, Saltyte Benth J, Bjorn Russell M. Validation of Placebo in a Manual Therapy Randomized Controlled Trial. Sci Rep 2015;5:11774. doi: 10.1038/srep11774 [published Online First: 2015/07/07]

35. Hohenschurz-Schmidt D, Draper-Rodi J, Vase L, et al. Blinding and sham control methods in trials of physical, psychological, and self-management interventions for pain (article II): a meta-analysis relating methods to trial results. PAIN 2022:10.1097.

36. Gevers-Montoro C, Provencher B, Descarreaux M, et al. Neurophysiological mechanisms of chiropractic spinal manipulation for spine pain. Eur J Pain 2021;25(7):1429-48. doi: 10.1002/ejp.1773 [published Online First: 2021/04/01]

37. Rubinstein SM, de Zoete A, van Middelkoop M, et al. Benefits and harms of spinal manipulative therapy for the treatment of chronic low back pain: systematic review and meta-analysis of randomised controlled trials. BMJ 2019;364:I689. doi: 10.1136/bmj.I689 [published Online First: 2019/03/15]

VERSION 2 – REVIEW

REVIEWER	Glissmann Nim, Casper
	University Hospital of Southern Denmark
REVIEW RETURNED	14-Nov-2022
GENERAL COMMENTS	I thank the authors for providing specific responses and adopting the
	manuscript accordingly. I have nothing further to add but to look

forward to the results.